Author's response to reviews

Title: Adhesive Capsulitis and Dynamic Splinting: a Controlled, Cohort Study

Authors:

Paul D Gaspar (gasparpt@hotmail.com)
F Buck Willis (Buckphd@yahoo.com)

Version: 4 Date: 12 August 2009

Author's response to reviews: see over
Response to Reviewer #1 (Nicola Massy-Westropp) 10 August 2009

Reviewer’s Report:
   6) The limitations of this study were previously described and can be read in lines 174-179.

Major Compulsory Revisions
   1) Due to the low power between groups, the post-hoc t-tests did not show a significant difference. This section has been re-written showing all the P and T values for each group. See lines 145-150.
   2) The “Intention to Treat” analysis is described in lines 131-134.
   3) Figure 2 now includes the wording “degrees of change in active range of motion, supine external rotation (humerus abducted to 90°)” so that this figure may now stand alone.

Thank you for the constructive review which will help benefit more patients.